Author's response to reviews

Title: Selenium enrichment of broccoli sprout extract increases chemosensitivity and apoptosis of LNCaP prostate cancer cells

Authors:

Rizky Abdulah (rizky@med.gunma-u.ac.jp)
Ahmad Faried (afaried@med.gunma-u.ac.jp)
Kenji Kobayashi (kenji-k@med.gunma-u.ac.jp)
Chiho Yamazaki (kchiho@health.gunma-u.ac.jp)
Eka W Suradji (suradji@med.gunma-u.ac.jp)
Kazuto Ito (kzito@med.gunma-u.ac.jp)
Kazuhiro Suzuki (kazu@showa.gunma-u.ac.jp)
Masami Murakami (mmurakam@showa.gunma-u.ac.jp)
Hiroyuki Kuwano (hkuwano@med.gunma-u.ac.jp)
Hiroshi Koyama (hkoyama@health.gunma-u.ac.jp)

Version: 4 Date: 6 November 2009

Author's response to reviews:

Dear Sir,

Thank you very much for accepting or manuscript. We have modified the format of the manuscript based on your instruction. Here we also add a second affiliation to some of the authors.

Sincerely,

Authors